Author’s response to reviews

Title: Anti-stress and neuronal cell differentiation induction effects of Rosmarinus officinalis L. essential oil

Authors:

Myra Villareal (villareal.myra.o.gn@u.tsukuba.ac.jp)
Ayumi Ikeya (ikeya327@gmail.com)
Kazunori Sasaki (sasaki-kazu@aist.go.jp)
Ben Arfa Abdelkarim (benarfaa200@yahoo.fr)
Mohamed Neffati (neffatimohamed@ira.rnrt.tn)
Hiroko Isoda (isoda.hiroko.ga@u.tsukuba.ac.jp)

Version: 2 Date: 11 Dec 2017

Author’s response to reviews:

Reply to the reviewers' comments:

Please make the following revisions to your manuscript:

1. Please include the email addresses of all authors on the title page of your manuscript.
   
   **REPLY:**
   
   We added the e-mail addresses of all the authors on the title page of the manuscript.

2. Please detail in your Methods section the euthanization method used in your study.
   
   **REPLY:**
   
   We added the details of the euthanization method used in the study (Page 7).